

January 2, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 14745-review.doc).

**Title:** New endoscopic ultrasound techniques: a comprehensive review

**Author:** Fansheng Meng, Zhaohong Zhang, Feng Ji

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 14745

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revisions have been made according to the suggestions of the reviewers

- (1) The manuscript has been reduced significantly;
- (2) The price and cost-effectiveness has been discussed;
- (3) The learning curve has been mentioned in the revised manuscript;
- (4) The development of EUS has been introduced in the revised manuscript.
- (5) The language has been polished by NPG Language Editing service.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Feng Ji, MD

Department of Gastroenterology

The First Affiliated Hospital

Zhejiang University, Qingchun Road No.79

Hangzhou 310000, China

E-mail: [jifeng1126@sina.com](mailto:jifeng1126@sina.com)